Author's response to reviews

Title: Intussusception of the small bowel in the elderly secondary to malignant metastases: a report of 2 cases.

Authors:

Charalambos Spiridis (cspyridi@auth.gr)
Apostolos Kambaroudis (kambarou@med.auth.gr)
Achilleas Ntinas (achippo@hotmail.com)
Savvas Papadopoulos (savvas_pap@hotmail.com)
Athanasios Papanikolaou (pap25@otenet.gr)
Thomas Gerasimidis (gerasimi@med.auth.gr)

Version: 4 Date: 21 November 2010

Author's response to reviews: see over
Dear Sir/Madame,
Thank you for your time and effort in examining this manuscript. All suggestions were acted upon.

Specifically,

1. Author list: corrected according to system entries.

2. References: Reference numbers were put in square brackets, and minor corrections done.

3. Consent: Consent section modified to indicate that both patients gave consent.

4. Authors Contributions: section formatted into a paragraph, and statement that all authors read and approved the final manuscript added.

5. Figure legends: Each figure was given a separate legend.

6. All highlighting was removed.

7. Title corrected to finish “2 case reports”.

8. Affiliations section corrected.

9. Corresponding author indicated.

We are at your disposal for any further corrections/modifications necessary.

Yours truly,
Charalambos Spiridis,  
Apostolos Kambaroudis,  
Achilleas Ntinas,  
Savvas Papadopoulos,  
Athanasios Papanicolaou,  
Thomas Gerasimidis